Author's response to reviews

Title: Impact of ambient air pollution on gestational age is modified by season in Sydney, Australia

Authors:

Bin Jalaludin (b.jalaludin@unsw.edu.au)
Trish Mannes (trish.mannes@email.cs.nsw.gov.au)
Geoffrey Morgan (gmorg@nrahs.nsw.gov.au)
Doug Lincoln (doug1@nrahs.nsw.gov.au)
Vicky Sheppeard (vshep@doh.health.nsw.gov.au)
Stephen Corbett (stephen_corbett@wsahs.nsw.gov.au)

Version: 5 Date: 28 May 2007

Author's response to reviews: see over
28 May 2007

Dr D Ozonoff
The Editor
*Environmental Health*

Dear Dr Ozonoff

**Re: Further revisions to manuscript titled “Impact of ambient air pollution on gestational age is modified by season in Sydney, Australia”**

Thank you for accepting our manuscript for publication following our responses to the Reviewers’ comments.

As requested by the Editorial Production Team we have now indicated that all authors have read and approved the final manuscript. We have also proof read the paper and made two minor changes.

We will be happy to respond to any further queries from the Editorial Production Team.

Yours sincerely

A/Prof Bin Jalaludin
Centre for Research, Evidence Management and Surveillance
Sydney South West Area Health Service
Locked Bag 7017
Liverpool NSW 1871
Australia

Tel: (+612) 9828-6000
Fax: (+612) 9828-6012
Email: b.jalaludin@unsw.edu.au